Author's response to reviews

Title: Preferences for Treatment of Attention-Deficit/Hyperactivity Disorder (ADHD): A Discrete Choice Experiment

Authors:

Axel C. Mühlbacher (muehlbacher@hs-nb.de)
Ina Rudolph (irudolph@its.jnj.com)
Hans-Joachim Linke (lincke@empirische-beratung.de)
Matthias Nübling (nuebling@empirische-beratung.de)

Version: 7 Date: 20 July 2009

Author's response to reviews: see over
Letter to the Editor

Title: Preferences for Treatment of Attention-Deficit/Hyperactivity Disorder (ADHD): A Discrete Choice Experiment

Manuscript ID 1692759872235312

Authors: Axel C. Mühlbacher, Ina Rudolph, Hans-Joachim Lincke, Matthias Nübling

Date: 18.07.2009

We went through the manuscript-formatting checklist and are sure that the revised manuscript conforms to all of the points.

We explained within the Methods section of our manuscript why ethical approval for your study was not needed.

„The study is a social science survey. It does not contain personal data (completely anonymous survey), surgeries (tests, experiments, medication), biomedical research or additional data, like in many epidemiological investigations. Therefore an ethic vote was not necessary. Study goals were explained to all participants and all gave written informed consent for their participation.“

Page 6 of the manuscript

Best Regards
Axel Mühlbacher